%====================================================================%
%                  Quark Matter 2012                                 %
% This latex file is a shortened version of the Elsevier template    %
% Please do not modify lines that have a comment ``do not change''   %
% associated with them. Those lines are needed for correct processing%
%====================================================================%

%% QM09 recommendation: please don't change layout or default packages
\documentclass[final,1p,times]{elsarticle} % do not change
%% for including figures
\usepackage{graphicx} % do not change
%% The amssymb package provides various useful mathematical symbols
\usepackage{amssymb} % do not change
%% The amsthm package provides extended theorem environments
\usepackage{amsthm} % do not change
%% The lineno packages: for possibly adding line numbers.
\usepackage{lineno} % do not change

\journal{Nuclear Physics A} % do not change
\begin{document}
\begin{thebibliography}{00} % do not change 

\bibitem{Dumitru:2002qt} A.~Dumitru, J.~Jalilian-Marian,
%``Forward quark jets from protons shattering the colored glass,''
{\it Phys.\ Rev.\ Lett.\ } \textbf{89}, 022301 (2002). %  [hep-ph/0204028].


%\cite{Kharzeev:2003wz}
\bibitem{Kharzeev:2003wz}
  D.~Kharzeev, Y.~V.~Kovchegov and K.~Tuchin,
  %``Cronin effect and high p(T) suppression in pA collisions,''
  {\it Phys.\ Rev.\  D} {\bf 68}, 094013 (2003);
%  [arXiv:hep-ph/0307037];
  %%CITATION = PHRVA,D68,094013;%%
  D.~Kharzeev, Y.~V.~Kovchegov and K.~Tuchin,
  %``Nuclear modification factor in d+Au collisions: Onset of suppression in the
  %color glass condensate,''
  {\it Phys.\ Lett.\  B} {\bf 599}, 23 (2004).
%  [arXiv:hep-ph/0405045].
  %%CITATION = PHLTA,B599,23;%%

%\cite{Albacete:2003iq}
\bibitem{Albacete:2003iq}
  J.~L.~Albacete, N.~Armesto, A.~Kovner, C.~A.~Salgado and U.~A.~Wiedemann,
  %``Energy dependence of the Cronin effect from nonlinear QCD evolution,''
  {\it Phys.\ Rev.\ Lett.\ } {\bf 92}, 082001 (2004).
 % [arXiv:hep-ph/0307179].
  %%CITATION = PRLTA,92,082001;%%

%\cite{Blaizot:2004wu}
\bibitem{Blaizot:2004wu}
  J.~P.~Blaizot, F.~Gelis and R.~Venugopalan,
  %``High-energy pA collisions in the color glass condensate approach. 1. Gluon production and the Cronin effect,''
  {\it Nucl.\ Phys.\ A} {\bf 743}, 13 (2004);
%  [hep-ph/0402256].
  %%CITATION = HEP-PH/0402256;%%\cite{Blaizot:2004wv}
%\bibitem{Blaizot:2004wv} J.~P.~Blaizot, F.~Gelis, R.~Venugopalan,
%``High-energy pA collisions in the color glass condensate approach. 2. Quark production,''
{\it Nucl.\ Phys.\ A} \textbf{743}, 57 (2004). %  [hep-ph/0402257].


%\cite{Dumitru:2005gt}

\bibitem{Dumitru:2005gt} A.~Dumitru, A.~Hayashigaki, J.~Jalilian-Marian,
%``The Color glass condensate and hadron production in the forward region,''
{\it Nucl.\ Phys.\ A} {\bf 765}, 464 (2006). %  [hep-ph/0506308].


%\cite{Baier:2005dz}
\bibitem{Baier:2005dz}
  R.~Baier, Y.~Mehtar-Tani and D.~Schiff,
  %``Has saturation physics been observed in deuteron-gold collisions at
  %RHIC?,''
  {\it Nucl.\ Phys.\  A} {\bf 764}, 515 (2006).
 % [arXiv:hep-ph/0508026].
  %%CITATION = NUPHA,A764,515;%%

\bibitem{JalilianMarian:2005jf}
J.~Jalilian-Marian and Y.~V.~Kovchegov,
%``Saturation physics and deuteron gold collisions at RHIC,''
{\it Prog.\ Part.\ Nucl.\ Phys.\ } \textbf{56}, 104 (2006).
%  [arXiv:hep-ph/0505052].
%%CITATION = PPNPD,56,104;%%and references therein.
%  [arXiv:1001.1378 [hep-ph]].

%\cite{Albacete:2010bs}
\bibitem{Albacete:2010bs}
  J.~L.~Albacete and C.~Marquet,
  %``Single Inclusive Hadron Production at RHIC and the LHC from the Color Glass
  %Condensate,''
  {\it Phys.\ Lett.\  B} {\bf 687}, 174 (2010).
 % [arXiv:1001.1378 [hep-ph]].
  %%CITATION = PHLTA,B687,174;%%

%\cite{Altinoluk:2011qy}

\bibitem{Altinoluk:2011qy} T.~Altinoluk, A.~Kovner,
%``Particle Production at High Energy and Large Transverse Momentum - 'The Hybrid Formalism' Revisited,''
{\it Phys.\ Rev.\ D} \textbf{83}, 105004 (2011). %  [arXiv:1102.5327 [hep-ph]].


%\cite{Qiu:2004da}
\bibitem{Qiu:2004da}
  J.~W.~Qiu and I.~Vitev,
  %``Coherent QCD multiple scattering in proton-nucleus collisions,''
  {\it Phys.\ Lett.\  B} {\bf 632}, 507 (2006).
 % [arXiv:hep-ph/0405068].
  %%CITATION = PHLTA,B632,507;%%

  %\cite{Guzey:2004zp}
\bibitem{Guzey:2004zp}
  V.~Guzey, M.~Strikman and W.~Vogelsang,
  %``Observations on dA scattering at forward rapidities,''
  {\it Phys.\ Lett.\  B} {\bf 603}, 173 (2004).
%  [arXiv:hep-ph/0407201].
  %%CITATION = PHLTA,B603,173;%%

%\cite{Kopeliovich:2005ym}
\bibitem{Kopeliovich:2005ym}
  B.~Z.~Kopeliovich, J.~Nemchik, I.~K.~Potashnikova, M.~B.~Johnson and I.~Schmidt,
  %``Breakdown of QCD factorization at large Feynman x,''
  {\it Phys.\ Rev.\  C} {\bf 72}, 054606 (2005).
 % [arXiv:hep-ph/0501260].
  %%CITATION = PHRVA,C72,054606;%%

%\cite{Frankfurt:2007rn}
\bibitem{Frankfurt:2007rn}
  L.~Frankfurt and M.~Strikman,
  %``Energy losses in the black disc regime and correlation effects in the STAR
  %forward pion production in d Au collisions,''
  {\it Phys.\ Lett.\  B} {\bf 645}, 412 (2007).
 % [arXiv:nucl-th/0603049].
  %%CITATION = PHLTA,B645,412;%%

%\cite{Arsene:2004ux}
\bibitem{Arsene:2004ux}
  I.~Arsene {\it et al.}  [BRAHMS Collaboration],
  %``On the evolution of the nuclear modification factors with rapidity and
  %centrality in d + Au collisions at s(NN)**(1/2) = 200-GeV,''
  {\it Phys.\ Rev.\ Lett.\ }  {\bf 93}, 242303 (2004).
%  [arXiv:nucl-ex/0403005];

%\cite{Adams:2006uz}
\bibitem{Adams:2006uz}
  J.~Adams {\it et al.}  [STAR Collaboration],
  %``Forward neutral pion production in p+p and d+Au collisions at  s(NN)**(1/2)
  %= 200-GeV,''
  {\it Phys.\ Rev.\ Lett.\ } {\bf 97}, 152302 (2006).
%  [arXiv:nucl-ex/0602011].
  %%CITATION = PRLTA,97,152302;%%

%\cite{McLerran:2011zz}

\bibitem{Gribov:1984tu} L.~V.~Gribov, E.~M.~Levin and M.~G.~Ryskin,
%``Semihard Processes in QCD,''
{\it Phys.\ Rept.\ } \textbf{100}, 1 (1983). %%CITATION = PRPLC,100,1;%%

%\cite{Mueller:1985wy}

\bibitem{Mueller:1985wy} A.~H.~Mueller and J.~w.~Qiu,
%``Gluon Recombination and Shadowing at Small Values of x,''
{\it Nucl.\ Phys.\ B} \textbf{268}, 427 (1986). %%CITATION = NUPHA,B268,427;%%

%
\bibitem{McLerran:1993ni} L.~D.~McLerran and R.~Venugopalan,
%``Computing quark and gluon distribution functions for very large nuclei,''
{\it Phys.\ Rev.\ D} \textbf{49}, 2233 (1994); %  [arXiv:hep-ph/9309289].
%%CITATION = PHRVA,D49,2233;%%
%\cite{McLerran:1993ka}
%\bibitem{McLerran:1993ka}
% L.~D.~McLerran and R.~Venugopalan,
%``Gluon distribution functions for very large nuclei at small transverse
%momentum,''
{\it Phys.\ Rev.\ D} \textbf{49}, 3352 (1994). %  [arXiv:hep-ph/9311205].
%%CITATION = PHRVA,D49,3352;%%



\bibitem{Chirilli:2011km}  G.~A.~Chirilli, B.~-W.~Xiao and F.~Yuan,
%``One-loop Factorization for Inclusive Hadron Production in $pA$ Collisions in the Saturation Formalism,''
{\it Phys.\ Rev.\  Lett. } {\bf 108}, 122301 (2012); arXiv:1203.6139 [hep-ph].
%%CITATION = ARXIV:1112.1061;%%


%\cite{CU-TP-441a}

\bibitem{CU-TP-441a} A.~H.~Mueller,
%``Small x Behavior and Parton Saturation: A QCD Model,''
{\it Nucl.\ Phys.\ B\ } \textbf{335}, 115 (1990); %%CITATION = NUPHA,B335,115;%%
%\cite{SLAC-PUB-10047}
%\bibitem{SLAC-PUB-10047}
%A.~H.~Mueller,
%``Soft gluons in the infinite momentum wave function and the BFKL pomeron,''
{\it Nucl.\ Phys.\ B\ } \textbf{415}, 373 (1994). %%CITATION = NUPHA,B415,373;%%

%

\bibitem{Balitsky:1995ub} I.~Balitsky,
%``Operator expansion for high-energy scattering,''
{\it Nucl.\ Phys.\ B}\textbf{463}, 99-160 (1996). %  [hep-ph/9509348].


\bibitem{Kovchegov:1999yj} Y.~V.~Kovchegov,
%``Small x F(2) structure function of a nucleus including multiple pomeron exchanges,''
{\it Phys.\ Rev.\ D }\textbf{60}, 034008 (1999). %  [hep-ph/9901281].

\bibitem{Mueller:1999wm}
  A.~H.~Mueller,
  %``Parton saturation at small x and in large nuclei,''
  {\it Nucl.\ Phys.\ B} {\bf 558}, 285 (1999).
%  [hep-ph/9904404].
  %%CITATION = HEP-PH/9904404;%%


\bibitem{Dominguez:2010xd} F.~Dominguez, B.~W.~Xiao and F.~Yuan,
%``$k_t$-factorization for Hard Processes in Nuclei,''
{\it Phys.\ Rev.\ Lett.\ } \textbf{106}, 022301 (2011).
% [arXiv:1009.2141 [hep-ph]].
%%CITATION = PRLTA,106,022301;%%
%\cite{Dominguez:2011wm}

\bibitem{Dominguez:2011wm} F.~Dominguez, C.~Marquet, B.~W.~Xiao and F.~Yuan,
%``Universality of Unintegrated Gluon Distributions at small x,''
{\it Phys.\ Rev.\ D} \textbf{83}, 105005 (2011). %  [arXiv:1101.0715 [hep-ph]].
%%CITATION = PHRVA,D83,105005;%%




\bibitem{Kovchegov:2006vj} Y.~V.~Kovchegov and H.~Weigert,
%``Triumvirate of Running Couplings in Small-x Evolution,''
{\it Nucl.\ Phys.\ A} \textbf{784}, 188 (2007). %  [hep-ph/0609090].
%%CITATION = NUPHA,A784,188;%%

\bibitem{Balitsky:2008zza} I.~Balitsky and G.~A.~Chirilli,
%``Next-To-Leading Order Evolution Of Color Dipoles,''
{\it Phys.\ Rev.\ D} \textbf{77}, 014019 (2008). % [arXiv:0710.4330 [hep-ph]].
%%CITATION = PHRVA,D77,014019;%%


\bibitem{Mueller:2012bn} 
  A.~H.~Mueller and S.~Munier,
  %``$p_{perpendicular}$-Broadening and production processes versus dipole/quadrupole amplitudes at next-to-leading order,''
  arXiv:1206.1333 [hep-ph].
  %%CITATION = ARXIV:1206.1333;%%


\end{thebibliography}
 % do not change

\begin{frontmatter} % do not change


%% Title, author and address info here; please do not use footnotes

% Your Title - please insert
\title{The NLO inclusive forward hadron production in pA collisions}

% Principle author, and co-authors - please insert
\author{Giovanni Chirilli$^{a}$, Bo-Wen Xiao$^{b,c}$ and Feng Yuan$^{a}$}

% Address - please insert
\address[a]{Nuclear Science Division, Lawrence Berkeley National Laboratory, Berkeley, CA 94720, USA}
\address[b]{Institute of Particle Physics, Central China Normal University, Wuhan 430079, China.}
\address[c]{Department of Physics, Pennsylvania State University, University Park, PA 16802, USA}

\begin{abstract} % do not change
%% Text of abstract goes here - please insert

Recently, the one-loop factorization for inclusive hadron productions in pA collisions in the saturation formalism has been established via the complete next-to-leading order calculation. The differential cross section is written into a factorization form in the coordinate space at the next-to-leading order, while the naive form of the convolution in the transverse momentum space does not hold. The rapidity divergence with small-x dipole gluon distribution of the nucleus is factorized into the energy evolution of the dipole gluon distribution function, which is known as the Balitsky-Kovchegov equation. Furthermore, the collinear divergences associated with the incoming parton distribution of the nucleon and the outgoing fragmentation function of the final state hadron are factorized into the splittings of the associated parton distribution and fragmentation functions, which allows us to reproduce the well-known DGLAP equation. The hard coefficient function, which is finite and free of divergence of any kind, is evaluated at one-loop order. This result is important, not only for the phenomenological applications to the inclusive hadron production in p-A collisions at RHIC and future LHC experiment, but also for theoretically promoting the rigorous developments towards factorizations in small-x physics.

\end{abstract} % do not change

\end{frontmatter} % do not change

%% we keep linenumbers at least for initial version
%\linenumbers % do not change

%% start of main text - please insert. 

%%\section{}\label{}

\section{Introduction}
Small-$x$ physics provides us the dense parton densities at high energy limit and predicts the onset of the saturation phenomenon as a result of nonlinear dynamics. 
Inclusive hadron production in $pA$ collisions, which is considered as one of the most important processes to reveal and test the small-$x$ physics, has attracted much of
theoretical interests in recent years~\cite{Dumitru:2002qt,Kharzeev:2003wz,Albacete:2003iq,Blaizot:2004wu,Baier:2005dz,Dumitru:2005gt,JalilianMarian:2005jf,Albacete:2010bs,Altinoluk:2011qy,Qiu:2004da,Guzey:2004zp,Kopeliovich:2005ym,Frankfurt:2007rn}.
In particular, the suppression of hadron production in the forward $dAu$ scattering at RHIC observed in the
experiments~\cite{Arsene:2004ux,Adams:2006uz} has been regarded as one of the evidences
for the gluon saturation at small-$x$ in a large nucleus~\cite{JalilianMarian:2005jf,Albacete:2010bs}.
Saturation phenomenon at small-$x$ in nucleon and nucleus plays an important role
in high energy hadronic scattering~\cite{Gribov:1984tu,Mueller:1985wy,McLerran:1993ni}. 
In this proceeding for Quark Matter 2012, we summarize our results which calculate the one-loop perturbative
corrections. Our results, which is an important step toward a complete description of hadron
production in $pA$ collisions in the small-$x$ limit, have been published earlier in Ref.~\cite{Chirilli:2011km}.

\section{The one-loop calculation of inclusive hadron productions in $pA$ collisions}

Inclusive hadron production in $pA$ collisions,
\begin{equation}
p+A\to h+X \ ,\label{pA}
\end{equation}
can be viewed as a process where a parton from the nucleon
(with momentum $p$) scatters on the nucleus target
(with momentum $P_A$), and fragments into final state hadron with momentum $%
P_h$. In the dense medium of the large nucleus and at small-$x$,
the multiple interactions become important, and we need to perform
the relevant resummation to make the reliable predictions.
This is particularly important because the final
state parton is a colored object. Its interactions with
the nucleus target before it fragments into the hadron is crucial
to understand the nuclear effects in this process.
In our calculations, we follow the high energy factorization, also called
color-dipole or color-glass-condensate (CGC),
formalism~\cite{CU-TP-441a,Balitsky:1995ub,Kovchegov:1999yj,Mueller:1999wm}
to evaluate the above process up to one-loop order.
%We notice that alternative approaches have been proposed in the
%literature~\cite{Qiu:2004da,Guzey:2004zp,Kopeliovich:2005ym,Frankfurt:2007rn}
%to calculate the nuclear effects in this process.

According to our calculations, the QCD factorization formalism for the above
process reads as,
\begin{eqnarray}
\frac{d^3\sigma^{ p+A\to h+X}}{dyd^2p_\perp}&=&\sum\limits_a \int \frac{dz}{%
z^2}\frac{dx}{x} \xi xf_a(x,\mu) D_{h/c}(z,\mu)% \nonumber \\ 
%&&\times
 \int [dx_\perp]
S_{a,c}^Y([x_\perp])\mathcal{H}_{a\to
c}(\alpha_s,\xi,[x_\perp]\mu) \ , \label{fac}
\end{eqnarray}
where $\xi=\tau/xz$ with $\tau=p_\perp e^y/\sqrt{s}$, $y$ and
$p_\perp$ the rapidity and transverse momentum for the final state
hadron and $s$ the total center of mass energy square
$s=(p+P_A)^2$, respectively. Schematically, the incoming parton
described by the parton distribution $f_a(x)$ scatters off the
nuclear target represented by multiple-point correlation function
$S^Y([x_\perp])$, and fragments into the final state hadron defined
by the fragmentation function $D_{h/c}(z)$. All these quantities
have clear operator definitions in QCD. In particular,
$f_a(x)$ and $D_{h/c}(z)$ are collinear parton distribution and fragmentation
functions which only depend on the longitudinal momentum fraction $x$ of
the nucleon carried by the parton $a$, and the momentum fraction $z$ of
parton $c$ carried by the final state hadron $h$, respectively.
From the nucleus side, it is the multi-point correlation functions
denoted as $S_{a,c}^Y(x_\perp)$ (see the definitions below) that enters
in the factorization formula, depending on the flavor of the
incoming and outgoing partons and the gluon rapidity $Y$ associated with
the nucleus: $Y\approx \ln(1/x_g)$ with $x_g$ being longitudinal momentum
fraction.


%At the leading order, $S^Y$ represent the two-point functions, including
%the dipole gluon distribution functions in the elementary and
%adjoint representations for the quark and gluon initialed
%subprocesses, respectively. Higher order
%corrections will have terms that depend on the correlation
%functions beyond the simple two-point functions. Because of this
%reason, the integral $[dx_\perp]$ represents all possible
%integrals at the particular order.

\begin{figure}[htb]
\begin{center}
\includegraphics[width=10cm]{divergences2}
\end{center}
\vspace*{8pt}
%\vspace*{-0.5cm}
\caption{Illustrations of the kinematic region of various divergences appearing in NLO calculation.\label{f2}}
\end{figure}

To evaluate NLO corrections, we need to calculate the gluon radiation
contributions. At one-loop order, the gluon radiation will introduce
various divergences. The factorization formula in Eq.~(\ref{fac})
is to factorize these divergences into the relevant factors.

For example, there is the rapidity divergence associated with
$S^Y([x_\perp])$ when the radiated gluon is collinear to the target nucleus. 
The physical interpretation of the rapidity divergence subtraction is 
quite interesting. The soft gluon is emitted from the projectile proton 
with momentum $(1-\xi)p^+$, and it is easy to see that the rapidity of 
this soft gluon goes to $-\infty$ when $\xi\to 1$ since the radiated gluon is now in the region $k_g^-\gg k_g^+$. As a matter of fact, 
this soft gluon can be regarded as collinear to the target nucleus 
which is moving on the backward light cone with the rapidity 
close to $-\infty$ and $P_A^-\gg P_A^+$. Therefore, it is quite natural to renormalize 
this soft gluon into the gluon distribution function of the target 
nucleus through the BK evolution equation\cite{Balitsky:1995ub,Kovchegov:1999yj}. In addition, there will be collinear divergences associated with the incoming
parton distribution or the final state fragmentation functions when the radiated gluon is either collinear to the incoming parton or to the outgoing parton as illustrated in last two figures of Fig.~\ref{f2}. As the standard procedure, we use the dimensional regularization ($D=4-2\epsilon$)
and follow the $\overline{\mathrm{MS}}$ subtraction scheme, in order to
compute and move the collinear divergence into the DGLAP equation.

The leading order results have been calculated before, from which we have
\begin{equation}
\mathcal{H}_{2qq}^{(0)}=e^{-ik_\perp\cdot r_\perp} \delta(1-\xi)\
,
\end{equation}
where $k_\perp=p_\perp/z$ and $r_\perp=x_\perp-y_\perp$.
The NLO calculations are obtained by using the dijet results calculated in Ref.~\cite{Dominguez:2010xd,Dominguez:2011wm} and then integrating over the phase space of the unobserved parton. After removing all the divergences as illustrated in Fig.~\ref{f2}, we can reach finite hard coefficients for the production cross section. 
For the quark channel contribution: $qA\to q+X$, the above factorization
formula can be explicitly written as
\begin{eqnarray}
\frac{d^3\sigma ^{p+A\to h+X}}{dyd^2p_\perp}&=& \int \frac{dz}{z^2}\frac{dx}{x%
}\xi xq(x,\mu) D_{h/q}(z,\mu)\int \frac{d^2x_\perp
d^2y_\perp}{\left(2\pi\right)^2} \nonumber \\
&&\left\{S^{(2)}(x_\perp,y_\perp)
\left[ \mathcal{H}_{2qq}^{(0)}
+\frac{\alpha_s}{2\pi}\mathcal{H}_{2qq}^{(1)}\right]\right.\nonumber\\
&&\left.+
\int \frac{d^2b_\perp}{(2\pi)^2}
S^{(4)}(x_\perp,b_\perp,y_\perp)\frac{\alpha_s}{2\pi}
\mathcal{H}_{4qq}^{(1)}\right\}\ ,\label{qchannel}
\end{eqnarray}
up to one-loop order.
The two-point and four-point functions are defined as
\begin{eqnarray}
S^{(2)}(x_\perp,y_\perp)&=&\frac{1}{N_c}\langle{U}(x_\perp){U}^\dagger(y_\perp)\rangle_Y\\
S^{(4)}(x_\perp,b_\perp,y_\perp)&=&\frac{1}{N_c^2}\langle \mathrm{Tr}[{U}(x_\perp)%
{U}^\dagger(b_\perp)] \mathrm{Tr}[{U}(b_\perp){U}%
^\dagger(y_\perp)]\rangle_Y\ ,
\end{eqnarray}
with $N_c$ the number of color in QCD, and
$U(x_\perp)$ %=\mathcal{P}\exp\left\{ig_S\int_{-\infty}^{+\infty} \text{d}x^+\,T^cA_c^-(x^+,x_\perp)\right\}$ 
is the
Wilson line in the small-$x$ formalism.% with $A_c^-(x^+,x_\perp)$ being the gluon field solution of the classical Yang-Mills equation.

After subtracting the above divergences, we obtain the hard
coefficients at large $N_c$ limit
\begin{eqnarray}
\mathcal{H}_{2qq}^{(1)} &=& C_{F}\mathcal{P}_{qq}(\xi )\ln
\frac{c_0^2}{r_\perp^2\mu ^{2}}\left(e^{-ik_\perp\cdot
r_\perp}+\frac{1}{\xi ^{2}}e^{-i\frac{k_\perp}{\xi}\cdot
r_\perp}\right) \nonumber \\
&&-3C_F\delta(1-\xi)e^{-ik_\perp\cdot r_\perp}\ln
\frac{c_0^2}{r_\perp^2k^{2}_\perp}   , \label{h2qq}\\
\mathcal{H}_{4qq}^{(1)}&=&-4\pi N_{c}e^{-ik_\perp\cdot r_\perp}\left\{
e^{-i\frac{1-\xi}{\xi}k_\perp\cdot (x_\perp-b_{\perp})}
 \frac{1+\xi ^{2}}{\left( 1-\xi \right) _{+}}\frac{1}{\xi }\frac{x_\perp-b_\perp
}{\left( x_\perp-b_\perp\right) ^{2}}\cdot \frac{y_\perp-b_\perp}{\left( y_\perp-b_\perp\right) ^{2}}
\right. \nonumber \\
&& \left.
-\delta (1-\xi )\int_{0}^{1}d\xi ^{\prime }\frac{1+\xi ^{\prime 2}}{%
\left( 1-\xi ^{\prime }\right) _{+}}\right.\nonumber \\
&&\left.\times \left[\frac{e^{-i(1-\xi')k_\perp\cdot
(y_\perp-b_\perp)}}{(b_\perp-y_\perp)^2}-\delta^{(2)}(b_\perp-y_\perp)\int
d^2 r_\perp' \frac{e^{ik_\perp\cdot r_\perp'}}{r_\perp^{\prime
2}}\right]\right\},
\end{eqnarray}%
where $c_0=2e^{-\gamma_E}$ with $\gamma_E$ the Euler constant. These hard coefficients do not contain any divergence. The next-to-leading order differential cross section is written into a factorization form in the coordinate space, while It is straightforward to see that the naive form of the convolution in the transverse momentum space does not hold. Furthermore, as part of the consistency check, we find that the above result is in agreement with the collinear factorization result in the dilute limit in the forward pA collisions.
The calculations for all other partonic channels $q\to q g$, $g\to gg$ and $g\to q\bar q$
follow the same procedure, and the corresponding
hard coefficients are calculated up to one-loop order in Ref.~\cite{Chirilli:2011km}.

\section{Conclusion}
Our calculations, together with NLO DGLAP and NLO BK\cite{Kovchegov:2006vj,Balitsky:2008zza} evolution equations, as well as the one-loop running coupling correction, 
provide the complete formula for inclusive hadron production at NLO. The corrections to this NLO order cross section are either
of order $\alpha_s ^2$ or suppressed by $\frac{1}{N_c^2}$. The numerical implementation of the above NLO correction will help us understand the forward inclusive hadron production in pA collisions phenomenologically and make the important precision test of saturation physics possible.
In terms of resummation, we will be able to resum $\alpha_s(\alpha_s\ln k_\perp^2)^n$ 
and $\alpha_s(\alpha_s\ln 1/x)^n$ terms. The extra factor of $\alpha_s$ can either 
come from the hard factor, which is calculated in this manuscript, or arise from the 
NLO DGLAP and BK evolution equations. 
The complete study of various processes\cite{Mueller:2012bn} will provide us important  steps to understand factorizations and phenomenology in the small-$x$ physics. 

%% end of main text

%\section*{Acknowledgments} % please insert, comment out or delete if not needed
%This is where one places acknowledgments for funding bodies etc., if needed.
%For the large collaborations, this is listed once and for all, together with 
%the author lists etc. in the proceedings back-material.

 % do not change 
\end{document} % do not change 
\endinput % do not change 
%%
%% EOF
